Author's response to reviews

Title: Hydroxyethyl starch versus ringer solution in cardiopulmonary bypass prime solutions

Authors:

Osman Tiryakioglu (tiryaki64@hotmail.com)
Gurdeniz Yildiz (gdyildiz1971@hotmail.com)
Hakan Vural (ahvural@hotmail.com)
Tugrul Goncu (drgoncu@hotmail.com)
Ahmet Ozyazicioglu (violinhemet@hotmail.com)
Senol Yavuz (syavuz@ttnet.net.tr)

Version: 2 Date: 1 July 2008

Author's response to reviews:

The requested changes in the format were performed. The titles of the authors were removed, keywords were omitted, tables heading and legends were rearranged, and the references were re-formatted as requested.

All of the authors have read this final version of the manuscript and have approved.

Sincerely yours,
On behalf of the authors,
Osman Tiryakioglu, MD.